Author's response to reviews

Title: Associations between gestational anthropometry, maternal HIV, and fetal and early infancy growth in a prospective rural/semi-rural Tanzanian cohort, 2012-13

Authors:

Amanda L Wilkinson (alw252@cornell.edu)
Sarah H Pedersen (sp729@cornell.edu)
Mark Urassa (urassamark@yahoo.co.uk)
Denna Michael (dennamichael@gmail.com)
Jim Todd (jim.todd@lshtm.ac.uk)
Safari M Kinung’hi (kinunghi_csm@hotmail.com)
John Changalucha (jchangalucha@yahoo.com)
Joann M McDermid (jm3xc@virginia.edu)

Version: 4 Date: 14 October 2015

Author’s response to reviews: see over
Dear Pregnancy & Childbirth Editor,

Thank you for sending us reviewer feedback on October 11th, 2015 that is related to a revised manuscript that was submitted for review on July 15th, 2015. Please find our point-by-point response to reviewer concerns:

Reviewer 1: No concerns were raised; no further comment.

Reviewer 2:

Point 1: Manuscript text modified as suggested (Lines 352-5).
Point 2: Reviewer opinion; no further comment needed.
Point 3: Manuscript text modified as suggested (Lines 288-91).
Point 4: Manuscript text modified as suggested (Lines 303-4).

Sincerely,

Joann M. McDermid